Author’s response to reviews

Title: Effects of low-flow sevoflurane anesthesia on renal function in low birth weight infants

Authors:

na xing Dr (mzknana@126.com)
Xin Wei Dr (787611423@qq.com)
Yanzi Chang Dr (Minerva@163.com)
Yingying Du Dr (duyy87@126.com)
Wei Zhang Dr (15003995323@163.com)

Version: 2 Date: 30 April 2014

Author’s response to reviews:

I only revise signature of the first autor in the end of cover letter, thank you!